Author's response to reviews

Title: The nutrition-based comprehensive intervention study on childhood obesity in China (NISCOC): a randomised cluster controlled trial

Authors:

Yanping Li (liyanping72@yahoo.com)
Xiaoqi Hu (huxiaoqee@163.com)
Qian Zhang (zhangq-99@263.net)
Ailing Liu (liuailing72@yahoo.com)
Hongyun Fang (fanghy1982@126.com)
Linan Hao (hln918@sina.com)
Yifan Duan (mantou_727@163.com)
Haiquan Xu (xhaiquan@163.com)
Xianwen Shang (andy243@126.com)
Jun Ma (majunt@bjmu.edu.cn)
Guifa Xu (yyxu@sdu.edu.cn)
Lin Du (dulinaa@hotmail.com)
Ying Li (liying_helen@163.com)
Hongwei Guo (hwguo@shmu.edu.cn)
Tingyu Li (tyli@vip.sina.com)
Guansheng Ma (mags@chinacdc.net.cn)

Version: 6 Date: 28 April 2010

Author's response to reviews:

Cover page line 12: "institute" was changed to "school"
Page 15 line 17 (or line 6 from the bottom): "Liubo Li" was revised to "Liubai Li"
Page 15 line 21 (or line 2 from the bottom): "Weijia Zhao" was revised to "Weijia Liu", "Jiangang Wu" was revised to "Jiagang Wu"